

Author's response to reviews

Title: Searching for Management Approaches to Reduce HAI Transmission (SMART): A Study Protocol

Authors:

Ann McAlearney (ann.mcalearney@osumc.edu)

Jennifer Hefner (jennifer.hefner@osumc.edu)

Cynthia Sieck (cynthia.sieck@osumc.edu)

Daniel Walker (daniel.walker@osumc.edu)

Alison Aldrich (alison.aldrich@osumc.edu)

Lindsey Sova (lindsey.sova@osumc.edu)

Alice Gaughan (alice.gaughan@osumc.edu)

Caitlin Slevin (caitlin.slevin@osumc.edu)

Courtney Hebert (courtney.hebert@osumc.edu)

Erinn Hade (erinn.hade@osumc.edu)

Jacalyn Buck (jacalyn.buck@osumc.edu)

Michele Grove (michele.grove@osumc.edu)

Timothy Huerta (timothy.huerta@osumc.edu)

Version: 1 Date: 06 Jun 2017

Author's response to reviews:

We have revised our manuscript to reduce the word count to 5461 words, under the limit of 5500 words. A copy of the manuscript with tracked changes is also included with this submission.

Thanks so much for your additional consideration of our work.

Best,

Ann and Team